Author’s response to reviews

Title: A self-report of the Healer’s Art by junior doctors: Does the course have a lasting influence on personal experience of humanism, self-nurturing skills and medical counterculture?

Authors:

Chanakya Jaiswal (u4773863@anu.edu.au)
Katrina Anderson (k.anderson@anu.edu.au)
Emily Haesler (emily.haesler@anu.edu.au)

Version: 3 Date: 15 Nov 2019

Author’s response to reviews:

1. Remove supplementary file
   Please remove the supplementary “JAISWAL_Chankya_Healer’sArt_Comments-addressed_2.docx” from the manuscript, as these files are not required to be attached to the manuscript at this stage of the editorial process. This has been removed.

2. Clean manuscript
   At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. This has been done and a new version attached.